Author's response to reviews

Title: A taboo within a stigma? A qualitative study of managing incontinence with people with dementia living at home

Authors:

Vari M Drennan (v.drennan@sgul.kingston.ac.uk)
Laura Cole (laura.cole@sgul.kingston.ac.uk)
Steve Iliffe (s.iliffe@ucl.ac.uk)

Version: 4 Date: 11 November 2011

Author's response to reviews: see over
Dear Editor

Thank you for the opportunity to address the formatting problems and other omissions. They are detailed below in bold.

Yours sincerely

Vari Drennan

---------------------------------

Formatting changes to be made:

Major revisions (we require the author to make these changes)

Title: please ensure that the title is identical in the manuscript and the submission system  These are now identical

References: Please make sure references are cited in numerical order. Currently reference 26 is the last reference cited. References have been checked and one duplicate removed. 1 to 30 have been cited in the text. 26 is in the table. 27-30 are in the discussion and conclusion.

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Affiliations: Please include countries.  Countries are now included

Tables: Please remove vertical lines.  The left and right borders of the table have now been removed